Author’s response to reviews

Title: Mobile Health in the Management of Type 1 Diabetes

Authors:
Xuemei Wang (wangxm_zsu@163.com)
Wei Shu (shuvimmu@163.com)
Jian Du (jdu-sdu@163.com)
Maolin Du (dumaolin2016@163.com)
Peiyu Wang (wpeiyu138@163.com)
Mingming Xue (happybird-nmg@163.com)
Huiqiu Zheng (zhxqqjye1988@163.com)
Yufeng Jiang (jiangyufeng@medmail.com.cn)
Shaohua Yin (ykdyinsh@163.com)
Danyan Liang (ldy77671@163.com)
Ruiqi Wang (wangruiqigood123@163.com)
Lina Hou (1328588962@qq.com)

Version: 1 Date: 05 Jul 2018

Author’s response to reviews:

RE: BEND-D-18-00220R1, entitled "Mobile Health in the Management of Type 1 Diabetes, A systematic review"

Dear Editor:

We really appreciate your great effort in reviewing our manuscript and provide us an opportunity to improve the quality of our paper. According to the reviewers’ suggestions, we have carefully revised and improved our manuscript. We added additional instructions to the end of the article, and modified the name in the Authors’ Contributions. The appendix added PRISMA checklist. We have highlighted the changes in the manuscript.
Your comments and suggestions really helped us a lot. We have put great efforts on them. We wish it can be satisfactory. If there is any information I can provide, please don’t hesitate to contact me.

Thank you again for your time and patience. Look forward to hearing from you.

Sincerely yours,

Maolin